Author's response to reviews

Title: Fecal pancreatic elastase-1 levels in older individuals without known gastrointestinal diseases or diabetes mellitus

Authors: Karl-Heinz Herzig (karl-heinz.herzig@oulu.fi)  
Anna-Kaisa Purhonen (anna-kaisa.purhonen@uku.fi)  
Kati M Räsänen (kati.m.rasanen@jyu.fi)  
Joanna Idziak (joanna.idziak@yahoo.com)  
Petri Juvonen (petri.juvonen@kuh.fi)  
Ryszard Philipps (rysphil@wp.pl)  
Jaroslaw Walkowiak (jarwalk@am.poznan.pl)

Version: 6 Date: 23 January 2011

Author's response to reviews:

Dear Production team,

we went through the manuscript again - please find the final version. We also uploaded it and uploaded new Fig 1 and 2 in addition. We now hope that the manuscript is ready for dissemination.

With kind regards

Karl-Heinz Herzig
(for the authors)